Author’s response to reviews

Title: Cost effectiveness of a multi-stage return to work program for workers on sick leave due to low back pain, design of a population based controlled trial.

Authors:

Mr Ivan A Steenstra (i.steenstra.emgo@med.vu.nl)
Johannes R Anema (h.anema@arbeid.tno.nl)
Paulien M Bongers (p.bongers@arbeid.tno.nl)
Prof Henrica CW De Vet (hcw.de_vet.emgo@med.vu.nl)
Willem Van Mechelen (w.vanmechelen@vumc.nl)

Version: 4 Date: 20 Oct 2003

L.s,

I have altered the manuscript according to the notes made by mr Krause. I have also made adjustments as requested by the editors. Unfortunately extraction of references didn’t work out the way it was supposed to. I've send an e-mail as requested.

with kind regards,

Ivan Steenstra